Author’s response to reviews

Title: Histological diagnosis of immune checkpoint inhibitor induced acute renal injury in patients with metastatic melanoma: a retrospective case series report

Authors:

Sebastian Hultin (sebastian.hultin@doctors.org.uk)
Kazi Nahar (kazi.nahar@melanoma.org.au)
Alexander M Menzies (alexander.menzies@sydney.edu.au)
Georgina V Long (georgina.long@melanoma.org.au)
Suran L. Fernando (suran.fernando@health.nsw.gov.au)
Victoria Atkinson (victoria.atkinson@health.qld.gov.au)
Jonathan Cebon (j.cebon@onjcri.org.au)
Muh Geot Wong (muhgeot.wong@sydney.edu.au)

Version: 1 Date: 15 Jun 2020

Author’s response to reviews:

1. Declaration page added as a separate heading and moved before references.
2. Title amended to reflect retrospective nature of study
3. Table 1 modified to include range of age of participants for purposes of patient anonymity.
4. Previously titled conclusions has been split into discussion and conclusion headings
5. Abbreviations and keywords moved after conclusions in main body and abstract respectively
6. Financial/funding disclosure renamed to funding